Author's response to reviews

Title: Weight change and incident metabolic syndrome in Iranian men and women; a 3 year follow-up study

Authors:

Azadeh Zabetian (zabetian@endocrine.ac.ir)
Farzad Hadaegh (fzhadaegh@endocrine.ac.ir)
Fateme Eskandari (fr-eskandari@yahoo.com)
Fereidoun Azizi (azizi@endocrine.ac.ir)

Version: 2 Date: 19 August 2008

Author's response to reviews: see over
Dear Melissa Norton
Editor-in-Chief,

We would like to apologize for revising submission. Since one word was missed in the title, we had to correct this point and besides we made two other typing corrections in the abstract and references. Once again we would like to thank you for your consideration of this manuscript and looking forward to hearing from you.

Dr. F. Hadaegh
Associate professor of Internal Medicine and Endocrinology
Prevention of Metabolic Disorders Research Center,
Research Institute for Endocrine Sciences,
Shaheed Beheshti University of Medical Sciences.
P.O. Box: 19395-4763, Tehran, Islamic Republic of Iran
Phone: 98 21 22432503
Fax: 22402463
Email: fzhadaegh@endocrine.ac.ir